%
% syllabus.tex
%

\documentclass{article}
\pagestyle{empty}

% Make left margins 1.25 inches
\setlength{\oddsidemargin}{.25in}
\setlength{\evensidemargin}{.25in}

% Line of text is 6 inches wide => right margin is 1.25 inches
\setlength{\textwidth}{6in}

% Make top margin 1 inch
\setlength{\topmargin}{0in}
\setlength{\headheight}{0in}
\setlength{\headsep}{0in}

% Length of body of text is 9 inches => bottom margin is 1 inch
\setlength{\textheight}{9in}

\setlength{\parindent}{0in}

\begin{document}

\begin{center}
\bf\LARGE Project Based Math 112, Fall 2001\\Syllabus
\end{center}

\begin{tabular}{ll}
\hspace{1.55in} & \\
week 1 (8/30--8/31) & intro \\[6pt]
week 2 (9/3--9/7) & \S 5.1 Areas \& distances \\
        & \S 5.2 The definite integral \\
        & \S 5.3 Evaluating definite integrals \\[6pt]
week 3 (9/10--9/14)
        & \S 5.4 The Fundamental Theorem of Calculus \\
        & \S 5.5 The substitution rule \\
        & \S 6.1 More about areas \\[6pt]
	& {\it project \#1 assigned (thurs/fri)} \\
week 4 (9/17--9/21)	
	& \S 6.2 Volumes \\
	& \S 5.6 Integration by parts \\
	& \S 5.7 Additional techniques of integration \\[6pt]
week 5 (9/25--9/28) & \S 5.9 Approximate integration \\
	& \S 5.10 Improper integrals \\
	& {\it project \#1 due (thurs/fri)} \\[6pt]
week 6 (10/1--10/5) & review for {\sc Prelim 1} (10/2) \\
	& \S 6.3  Arc length \\
	& \S 6.4 Average value of a function \\[6pt]
week 7 (10/8--10/12) & {\sc Fall Break} \\
	& \S 6.7 Probability \\
	& \S 8.1 Sequences \\[6pt]
week 8 (10/15--10/19) & \S 8.1 Sequences \\
	& \S 8.2 Series \\
	& {\it project \#2 assigned (wed)} \\[6pt]
week 9 (10/22--10/26)	
	& \S 8.3 The integral and comparison tests; estimating sums \\
	& \S 8.4 Other convergence tests \\[6pt]
week 10 (10/29--11/2) 	
	& review for {\sc Prelim 2} (10/30) \\
	& \S 8.5 Power series \\
	& \S 8.6 Representations of functions as power series \\[6pt]
week 11 (11/5--11/9) & {\it project \#2 due (mon)} \\
	& \S 8.6 Representations of functions as power series \\
	& \S 8.7 Taylor \& Maclaurin series \\[6pt]
week 12 (11/12--11/16)
	& \S 7.1 Modeling with differential equations \\
	& \S 7.2 Direction fields and Euler's methods \\
	& {\it project \#3 assigned} \\
	& \S 7.3 Separable equations \\[6pt]
week 13 (11/19--11/23) & \S 7.4 Exponential growth and decay \\
	& \S 7.5 The logistic equation \\
	& {\sc Thanksgiving} \\[6pt]
week 14 (11/26--11/30)
	& \S 8.10 Using series to solve differential equations \\[6pt]
week 15 (12/3--12/7) & catch up and review \\
	& {\it project \#3 due} \\
	& {\sc Final Exam:} Thursday 12/13 \\
\end{tabular}

\end{document}
